Author's response to reviews

Title: Effectiveness of a PLISSIT model intervention in patients with type 2 diabetes mellitus in primary care: design of a cluster-randomised controlled trial

Authors:

Anne Rutte (a.rutte@vumc.nl)
Patricia Van Oppen (p.vanoppen@ggzingeest.nl)
Giel Nijpels (g.nijpels@vumc.nl)
Frank J Snoek (fj.snoek@vumc.nl)
Paul Enzlin (paul.enzlin@uzleuven.be)
Peter Leusink (leusinkp@knmg.nl)
Petra JM Elders (p.elders@vumc.nl)

Version: 3
Date: 18 May 2015

Author's response to reviews: see over
Dear Sir/Madam,

We would like to submit a revision of our study protocol titled ‘Effectiveness of a PLISSIT model intervention in patients with type 2 diabetes mellitus in primary care: design of a cluster-randomised controlled trial’ for publication in *BMC Family Practice* (Manuscript number 1423379114168325).

As requested by Ms. Valenzuela, we made the following changes to the original manuscript:

- Page 5, line 28-29: “The study protocol was peer-reviewed by the Dutch Diabetes Research Foundation.” was added to the Methods/Design section to confirm that our study protocol underwent peer-review by the funding organization.
- Page 2, line 8: the word count was changed from 3811 to 3822.

Hoping to have informed you sufficiently,
Yours sincerely,
Anne Rutte